Author's response to reviews

Title: Molecular genetics analysis of hereditary breast and ovarian cancer patients in India

Authors:

NAGASAMY SOUMITTRA (soumi_n@yahoo.co.uk)
BALAIAH MEENAKUMARI (hema_meenakumari@yahoo.co.in)
TITHI PARIJA (tithiparija@yahoo.com)
VELUSWAMI SRIDEVI (sri04devi@yahoo.co.in)
KARUNAKARAN N NANCY (nirc18@gmail.com)
RAJARAMAN SWAMINATHAN (rcsurvival@yahoo.co.uk)
KAMALALAYAM R RAJALEKSHMY (rajalekshmy_raghavan@yahoo.com)
URMILA MAJHI (umajhi@hotmail.com)
THANGARAJAN RAJKUMAR (drtrajkumar@gmail.com)

Version: 5 Date: 1 August 2009

Author's response to reviews:

01-08-09
Sir,
I am herewith enclosing the revised manuscript. It has the following revisions
1. The Authors contributions has been corrected as advised.
2. The references have been corrected.
3. Authors names are not in capitals in the submission system
4. Key words have been removed.
5. Figure numbers have been removed from the Figures
6. Title is in sentence case
7. Country has been mentioned in the affiliations.
8. The vertical lines in tables has been removed (appears a ghost lines)
Thanking you,
Yours sincerely,
Dr.T.Rajkumar